Author's response to reviews

**Title:** The precarious supply of physical therapists across Canada: Exploring national trends in health human resources (1991 to 2005)

**Authors:**

Michel D Landry (mike.landry@utoronto.ca)
Thomas C Ricketts (ricketts@schsr.unc.edu)
Molly C Verrier (m.verrier@utoronto.ca)

**Version:** 6 **Date:** 17 September 2007

**Author's response to reviews:** see over
September 17, 2007

RE: MS: 1416563957142330
Journal: Human Resources for Health
Authors: Michel D Landry, Thomas C Ricketts and Molly C Verrier
Title: The precarious supply of physical therapists across Canada: Exploring national trends in health human resources (1991 to 2005)

Dear Editors:

Thank you again for the opportunity to publish our work in Human Resources for Health. This letter outlines the changes that we have made based on requests made from an email dated September 14th, 2007.

1. References: The references in this manuscript were performed manually. We would be pleased to pay the £30 from the journal to format these references. Please supply the correct information and we will pay this cost.

2. Figures. As per your request, we have deleted images from the text file of the manuscript, and we have moved the legend for figure 2 to below the title for figure 2.

3. Key words from the abstract.

4. Tables. The source of information from the tables has been deleted. We have cited the correct references for each table. We did not have to add references, as the source was already in the reference list.

5. Affiliations. The name of country was added to each of the affiliations.

Best regards,

Mike

Michel D. Landry PT, PhD
Assistant Professor, Department of Physical Therapy,
University of Toronto, 160-500 University Avenue
Toronto, Ontario, Canada, M5G 1V7
Tel: 416-946-8560
Fax: 416-946-8562
Email: mike.landry@utoronto.ca